Author's response to reviews

Title: A Rare Cause Of Ascites: a case report

Authors:

Iftkhar Ahmed (ansari35@yahoo.com)
Wong C Tan (chew.tan@nch.nhs.uk)
Anastasios Koulouzidis (akoulouzidis@hotmail.com)
Javaid Iqbal (doctor_javaid@hotmail.com)

Version: 6 Date: 15 November 2007

Author's response to reviews:

Referee 1 Dr. Manu Nayar

1) history of exposure to asbestos and family history and mesothelioma?

I have provided this information that there was no past exposure of asbestos and no family history of mesothelioma.

2) to shorten the case report and discussion

I have tried to make the report as concise as possible and of course discussion is already very brief and to shorten it further will leave many issue unanswered.

3) only one picture is enough

I leave it with you if you put all pictures or put one for CT and one for Histology.

Dr. Dimitrios Filippou

Explanation of further treatment?

I have added brief about the further treatment of the patient and of course its details are unnecessary.

Referee 3 hasn't made any comments and agreed for its publication. In fact all referees agreed that this case is well written and fulfill criteria to be published which is really encouraging.

If you need any more information please do not hesitate to contact me.

Let me take this opportunity to thank you for all your help and support.

Kinds Regards,

Dr. iftikhar Ahmed